Author's response to reviews

Title: New evidence on the process of colonic anastomosis healing in Wistar rats

Authors:

Christoforos Kosmidis (dr.ckosmidis@gmail.com)
Christoforos Efthimiadis (cevis@otenet.gr)
Georgios Anthimidis (anthe_geo@mail.com)
George Basbanis (gbasdanis@gmail.com)
Stelios Apostolidis (stel.apostolidis@yahoo.gr)
Prodromos Chitirog lou (p.chitirog lou@yahoo.gr)
Epameinondas Fahantidis (mennipur_cynic@yahoo.gr)

Version: 2 Date: 23 March 2010

Author's response to reviews: see over
To
The Editorial Office
«BMC Surgery»

Dear Sir,

We are submitting a copy of the brief clinical report entitled «New evidence on the process of colonic anastomosis healing in Wistar rats». We would appreciate it if you could consider the manuscript for publication in «BMC Surgery». We submit this paper for publication in your journal because this journal delivers up-to-date coverage of both basic and clinical medicine. Our study covers a topic relevant to basic and clinical area of interest, as it elucidates and adds new evidence on the events responsible for failure of colonic anastomosis healing.

Thank you in advance for your time and consideration.

We also state that:

I. The work is original, has not been accepted for publication nor is concurrently under consideration elsewhere, and will not be published elsewhere without the permission of the Editor.

II. All the authors have contributed directly to the planning, execution or analysis of the work reported or to the writing of the paper and there is no conflict to disclose.

III. No affiliations or financial involvement exist.

Sincerely yours,

Christophoros Kosmidis

MD, PhD